**Author's response to reviews**

**Title:** Hwangryunhaedoktang in adult patients with Atopic Dermatitis: a randomised, double-blind, placebo-controlled, two-centre trial - study protocol

**Authors:**

- Nam-Kwen Kim (drkim@wonkwang.ac.kr)
- Dong-Hyo Lee (secretop17@naver.com)
- Hyung-Sik Seo (aran99@naver.com)
- Seung-Ho Sun (sunguy2001@hanmail.net)
- Yong-Leol Oh (medoyl@dreamwiz.com)
- Ji-Eun Kim (kshijeun@hanmail.net)
- In-Hwan Yoon (andgarsia@hanmail.net)
- Eun-Sung Seo (drses@hanmail.net)
- Gye-Seon Shim (moonrise123@hanmail.net)
- Christopher Zaslawski (chris.zaslawski@uts.edu.au)

**Version:** 4  **Date:** 22 August 2011

**Author's response to reviews:**

Dear editors,

All authors appreciated the editor's comment for improving the manuscript. We have now cited the figure in the manuscript text as 'Figure 1'. We've highlighted the word with yellow color. We have placed the appendix before the references. Also we revised the 'figure 1' format. We carefully checked the manuscript for any typing or content errors.